To,

The Editor

World Journal of Gastroenterology

Subject: Revision of manuscript (Manuscript No: 87564)

Dear Sir,

This is in reference to your email dated August 25, 2023 (Subject: World Journal of Gastroenterology Manuscript NO: 87564 – Manuscript requires a revision) regarding the reviewers’ comments on the “Correction Note” submitted to your journal for the consideration. The reviewers’ comments are mentioned below:

**Reviewer #1:**

Scientific Quality: Grade A (Excellent)

Language Quality: Grade A (Priority publishing)

Conclusion: Accept (High priority)

Specific Comments to Authors: Accept as it is.

**Reviewer #2:**

Scientific Quality: Grade A (Excellent)

Language Quality: Grade A (Priority publishing)

Conclusion: Accept (High priority)

Specific Comments to Authors: The correction note is fine and the manuscript correction is to be published.
Since both the reviewers have not suggested any corrections or given any suggestions for the submitted “Correction Note”, we are uploading the “Correction Note” as it is through your online submission portal. We request you to publish the “correction note” as a “corrigendum” to our already published manuscript entitled “Role of Prebiotics, Probiotics, and Synbiotics in Management of Inflammatory Bowel Disease: Current Perspectives”.

Thanks & Regards

Prof. (Dr.) Suneela Dhaneshwar

Director, Amity Institute of Pharmacy

Amity University Maharashtra

Mumbai-Pune Expressway, Bhatan, Post-Somathne, Panvel, Mumbai 410206, Maharashtra, India.